

EMOTIONALLY WRAPPED SOCIAL MEDIA TEXT: APPROACHES, OPPORTUNITIES, AND CHALLENGES

TARA RAWAT *AND SHIKHA JAIN †

Abstract. With the evolution of several online platforms for information sharing such as social media, blogs, product review sites, and discussion forums, people have become more proactive in sharing their expectations, views, feelings, and experiences. This large amount of emotionally wrapped data motivates many researchers to perform data mining and present the crux of hidden emotions or mental states in a more presentable and comprehensible manner. It has several applications in different domains such as business, education, psychology, politics, and many more. This paper presents a detailed literature review projected to rigorously analyze the existing approaches to identify the mental or emotional state of a person from unstructured textual data. We include the most relevant papers which were published during 2001-2022. The selected papers are classified into three categories: granularity level, contextual level, and cognition level. Each category is carefully analyzed followed by a detailed and critical discussion. Finally, open challenges, opportunities, applications, and future directives are presented in-depth to facilitate the researchers working in the domain of emotion mining.

Key words: Emotion Mining; Opinion Mining; Sentiment Analysis; Social Media; Affective Computing; Emotion; Mentalstate

1. Introduction. The evolution of the web has given users the opportunities to participate, share, and contribute to various platforms like social media, blogs, product review websites, online forums, and discussion platforms. Now, it has been very common that as a kind of relief, people suffering from mental health issues frequently either directly or indirectly reveal their emotions, feelings and everyday battles with mental health concerns on social media ([6]. As a result, a large amount of multi-modal data is generated.

With the advancement of data, affective computing (AC) and emotion mining (EM) have emerged as new areas of research. Affective Computing (AC) is a field that relates to, arises from or deliberately influences emotion or other affective phenomena [97]. It plays a vital role in effective communication. It is broadly classified as emotion recognition, modeling, and expression (as shown in Fig. 1.1). Emotion recognition means extracting the emotional state from different modalities such as image, video, text, and audio-visual. Emotion modeling refers to the study of the effects of emotions on different cognitive processes and generating synthetic emotions. Emotion expression means expressing emotions using different modalities. It is important to mention that the researchers have used different vocabulary like sentiment, emotion, opinion, etc in the literature. According to Gordon [39], sentiments can be defined as a collectively formed set of neuronal responses, emotional attitudes, and common definitions normally coordinated around another individual. Emotion [124] has four interrelated components: sensory stimuli, physiological shifts, verbal movements, and an emotion tag which indicates the local correlation of the components.

In consideration of the similarities of how emotions and sentiments have been described, we will use these words interchangeably in this study to refer to perceptions that arise from the cumulative forces of the mental, behavioral, and personal [121] Many researchers have worked towards identifying the polarity of emotions in terms of negative, positive, or neutral emotions and called it as emotion mining [2],[83]. However, some researchers identified emotions in terms of discrete emotions such as happiness, sadness, fear, anger, and so [151]. Predominantly, both kind of work is called emotion mining. Emotion mining is one of the most challenging tasks to extract the emotions of people from the web, inferring from the text, and predicting the underlying intent. The eruption of research in emotion mining took place during 2001-2022, as shown in (Fig. 1.2). We

^{*}Jaypee Institute of Information Technology, Noida, India. (tararawat2503@gmail.com).

[†]Jaypee Institute of Information Technology, Noida, India. (shi_81@rediffmail.com).



Fig. 1.1: Affective Computing Classification



Fig. 1.2: Graph showing the research in "Emotion Mining from Text" in Google Scholar Database

sought to ensure the reliability and relevance of our dataset by implementing specific inclusion and exclusion criteria. The inclusion criteria are as follows:

- 1. Publications were included if they directly addressed the research topic of "Emotion Mining from Text" on the Google Scholar Database.
- 2. Only peer-reviewed journal articles and conference proceedings were considered to maintain the quality and reliability of the sources.
- 3. Publications within the time frame of 2001 to 2022 were included to focus on recent research trends.
- 4. Inclusion was limited to publications in English to facilitate data extraction and analysis.

The Exclusion Criteria:

- 1. Publications lacking titles, abstracts, or keywords directly related to the phrase "Emotion Mining from Text" research topic were excluded.
- 2. Non-peer-reviewed sources, such as books and theses, were excluded to ensure data quality.
- 3. Publications outside the specified date range 2001-2022 were excluded to maintain a focus on recent research.
- 4. Publications in languages other than English were excluded due to language proficiency constraints.

Since then, emotion mining has explored various domains such as E-therapy, psychological health services, counseling online, terrorist attacks [16], prediction of the financial market [56], medicine, and the healthcare domain [29], [110]. Recently, researchers [5], [45], [35] used a combination of cognitive and affective information with human language to tackle acute issues such as irony and sarcasm detection [80] The authors used crowd-sourcing [20] to build a common sense knowledge base, which is used to link the semantic gaps between

word-level and concept-level analysis. Furthermore, for context-aware and cognition-based emotion mining, strenuous attempts are advancing from polarity detection to more complex nuances of users' social behavior on online platforms [41], [60].

This paper presents a critical analysis of the work done by experts in the field of emotion mining while extracting emotional and mental states from textual data available on the web. The key features of the paper are:

- 1. An extensive literature survey of the most relevant papers published in the last two decades is presented to demonstrate the incremental approaches and improvements.
- 2. Categorical analysis (Fig. 2.1) and critical discussion of each category is presented.
- 3. Tabular summary of important papers is illustrated to provide a quick overview.
- 4. Generalized framework of emotion mining extracted from papers is presented to guide beginners in this field.
- 5. Over the year evolution of different techniques for emotion mining is presented pictorially to give a comprehensive view.
- 6. Various challenges encountered over the year are discussed in detail and presented pictorially.
- 7. Applications of emotion mining are discussed.
- 8. Overall discussion on EM with future directives is presented at the end of the paper.

The paper follows the incremental approach on the timeline of the last two decades. To the best of the authors' knowledge, no survey paper covers belief-based and semantic-based emotion mining which is the future of the field.

The rest of the paper is organized as follows: Section 2 reviews the advances in emotion mining and Sections 3,4 and Section 5 discusses various levels of text-based emotion mining. Section 6 summarizes the discussed approaches. Existing challenges, opportunities, and their applications are discussed in Sections 7,8, and Section 9 respectively. Finally, Section 10 concludes the paper and provides future directives.

2. Advances in Emotion Mining from Text. Textual emotion mining consists of collecting facts about beliefs, thoughts, and emotions that persons express about topics of interest. The field of emotion mining in computational linguistics focuses primarily on classifying different emotional contents within a word, phrase, or document. This field typically includes tasks such as defining emotions, classifying subjectivity, and understanding polarity [87] Analyzing the sentiment of textual data can, therefore, involve investigating the emotion/mental behind that text. In certain cases, emotion motivates an individual to evaluate an event and to create opinions about it. The field depicts a large problem space. There are a few terms that have mutual meaning such as sentiment analysis, opinion mining, emotion mining, and subjectivity analysis, affect analysis, review assessment, opinion extraction. However, all these terms come under emotion mining and sentiment mining are often employed [67] Hence, this paper explores all aspects of emotion mining (such as emotion analysis, emotion mining, opinion mining, etc.) termed differently in literature. These terms are used interchangeably throughout the paper to retain the original terminology used in different articles that are discussed here.

A range of focused and advanced fields are already being explored. However, text-based emotion mining still has a long way to go. Studying emotions in different fields provides us with useful information on what kind of emotions people experience while dealing with different situations.

This paper broadly investigates the literature under three categories: granularity level, cognition level, and contextual level. The same has been shown in (Fig. 2.1) and discussed in detail in subsequent sections.

3. Granularity-based ttextual Emotion Mining. In this section, we include a discussion on the related research for emotion mining based on granularity such as word level, sentence level, and document level.

3.1. Word-level Processing. Words may be demonstrative, expressive, and sensitive. Identifying the describing words (adjectives, adverbs, some nouns, and verbs) out of the whole text is the main focus at word-level emotion mining where each word is assigned a value on a scale from 0 (negative) to 10 (positive) based upon its polarity. SentiWordNet [31] is designed to annotate WordNet. It is used for calculating the score of a particular word in a sentence using machine learning approaches. For example, "This place is beautiful", this



Fig. 2.1: Classification criteria of Text-Based Emotion Mining

sentence has the word 'beautiful' that has some score as defined in lexicons, which leads to a positive polarity. Each word has some probability score for each emotion. The highest score is chosen as the emotion of that word. It has been observed that in natural language, opinionated content or the word, which expresses the emotion, is most commonly conveyed by certain parts of speech (POS). For instance, adjectives such as disastrous or lovely convey emotions more effectively than the verb running or eating. Hence, POS is used to mark words and search for polarity terms (adverbs and adjectives, [76]. Broadly, word-level EM can be done in three steps:

- 1. Detecting the subjectivity of different words. For example, love and hate, both are subjective words.
- 2. Calculating the polarity of words i.e. positive, negative, or neutral by using polarity scores. For instance, love is positive and hate is negative
- 3. Calculating the intensity of the word i.e. depth of positive, negative by analyzing the polarity words.

Opinion orientation can be determined by two approaches [3] a) Corpus-based b) Lexicon-based. The corpusbased approach detects opinion words with context-specific orientation [137] and lexicon-based approaches detect the seed words and search them with the associated lexicons such as WordNet along with their synonyms and antonyms [135]. Recently, domain-specific areas are explored rigorously by researchers. They have been working to increase the classifier's reliability by augmenting word-level emotive information, i.e., constructing and utilizing various forms of custom dictionaries [64].

3.2. Document-level Processing. At the document level, the aim is to analyze and classify the whole document as positive or negative. Several studies have been explored at the document level, especially, in cross-domain, cross-language, and opinion categorization [125], [27], [139]. Bellstam et al. [7], used the finance-related corpora, generated by financial analysts. They explored the emotions/sentiments in textual descriptions of business operations to quantify corporate innovation in finance.

Duyu et al. [123] proposed Convolution Neural Networks (CNN) and Recursive Neural Network (RNN) based document-level analysis [34], [133] called User Product Neural Network (UPNN). This model assimilates user and object-level information for sentiment classification for document-level analysis. They experimented with improved accuracy using vector-based and matrix-based user and object representation. Liu et al. [71] compared the sentiment classification approaches, using Recursive Neural Network and Recurrent Neural Network, and found that the sentiment topics or target entities cannot be identified by document or sentence level, hence need to analyze at a more granular level.

3.3. Sentence-level Processing. The goal of sentence-level emotion mining is to separate emotional information from facts to improve the prediction accuracy of the document's overall polarity [143]. At this level, the concern is basically on two steps:

- 1. The subjectivity of the sentence
- 2. Polarity (+ve or -ve) of the sentence

The following section will discuss the major input for both tasks.

3.3.1. Classifying Subjective Sentence. There are many approaches to classify a given sentence as subjective or objective. An objective sentence provides some factual information about the environment, while a subjective sentence communicates some personal emotions, perceptions, or opinions. Emotions are considered subjective feelings and thoughts [70], [144]. Some researchers feel that objective sentences do not imply emotions. However, they can also express emotions, and ignoring them may result in the wrong prediction. In this context, Liu [68] has discussed that subjectivity and opinion should not be compared. He also discussed that a subjective sentence may not convey any emotion such as "I am using the latest version of Android OS" is a subjective sentence, even then it doesn't express any emotion. Similarly, an objective sentence may imply emotion such as "After updating the latest android OS, my phone kept hanging". While this sentence is stating a fact, it shows an implicit emotion about the topic (the phone hanging).

Classification of subjective sentences is one of the most important and foundation schemes. The existence of subjective words determines the subjectivity of a sentence [42], [154], [47]. Moreover, these studies have been able to differentiate between subjective and objective sentences which subsequently improved sentence-level emotion mining. Several machine-learning approaches with language-specific features for Urdu languages are used [86] to identify the subjectivity of sentences. Therefore, different sentences may be dealt with differently at sentence-level analysis.

3.3.2. Polarity identification. The purpose of polarity identification is to determine text as positive, negative, or neutral according to its emotional significance. Some authors [1] the WordNetAffect Affective database [99] to distinguish between positive, negative, and neutral emotions. However, these emotional categories may help predict polarity, but not quite enough to predict the polarity intensity. Therefore, the authors use a few emotional categories such as love, happiness, joy, fear, sadness, anger, and so on. One approach is to use parts-of-speech (POS) tags syntactic patterns to express opinions/sentiments/emotions. The researchers [58] used the approach of multiplying all the scores of emotional words in the sentence as +1 and -1. They also used classification techniques to identify a particular type of emotional orientation. The hierarchical sequence learning model identifies polarity in both sentences as well as document-level and also improved the accuracy at both levels.

In 2010, Hassan et al. [4] proposed a method to identify online users' behavior using the Markov Model. Then, they identified the polarity orientation of the attitudes. Some researchers worked on target-dependent sentiment classification, where the target is classified into some category [51]. Researchers admitted that there are a variety of sentences and each sentence expresses emotions differently. Therefore, one technique cannot be applied to all kinds of sentences. Handling sarcastic sentences is still a big challenge as positive sentences may have negative meanings and vice versa. These types of sentences are more used in online discussions and political forums. Tsur et al. [127] proposed a semi-supervised learning approach to identify sarcastic and non-sarcastic sentences. Initially, they used the labeled set and expanded it automatically through a web search. Many researchers [66], [157] experimented with Convolution Neural Networks (CNN) for sentence-level classification. They built the model on top of the unsupervised pre-trained word vector, tuned hyper-parameters, and performed well with remarkably improved results.

3.4. Critical Analysis of Granularity-based Emotion Mining. Though a lot of research is conducted at a granular level, it has several limitations when used in real-life applications, such as which features or aspects of the entities are liked or disliked. Several sentences are not easily classified as emotion orientation varies on different targets or entities, e.g., "Trying out Airtel sim because Jio voice calls being cut after every 10 mins." and "Vodafone is doing well in this awful environment". In the latter sentence, sentence-level classification is insufficient. We need to go to the aspect-level analysis. It is assumed that a sentence has an overall positive or negative orientation but some of its parts may convey the opposite emotion. E.g., "India's labor-intensive agriculture has achieved steady increases in food grain production despite the often unfavorable weather conditions". Here, the overall message is positive but it also contains a negative emotion "unfavorable" which cannot be ignored and only aspect-level analysis can solve this problem. Moreover, comparative sentences

cannot be dealt with at the sentence level. For instance, the sentence "Amul butter tastes better than Mother Dairy butter" can't be simply classified as positive, negative, or neutral. Therefore, further, fine-grained analysis is required to handle them.

4. Context-based Textual Emotion Mining. Features are the attributes of an entity. Feature-based emotion mining extracts the emotions towards some features of the entity [115]. It also analyzes the emotion while considering the surrounding context of the feature. This section discusses the two types of feature-based emotion mining: aspect-level and context-aware.

4.1. Aspect-level Emotion Mining. The aspect-level analysis mainly focuses on entities and their components, where entities may be a product, an organization, a service, or a topic. The components can be referred to as features or attributes of the entity. Aspect-based emotion mining, also known as feature-based emotion mining, identifies different aspects of the target entity in a long sentence and its polarity [93]. Earlier, this kind of analysis required a set of features which are manually designed. Nowadays, the focus has shifted to other approaches such as graphs of dependency relationships [152], and lexicon-based supervised learning approach [55]. It is observed that identifying entities, their components, and knowing their polarity is not sufficient to know the sentiment of a sentence. For instance, a negative emotion about an entity does not mean that all the aspects of that entity will have a negative orientation [120].

The aspect-based emotion mining is carried out at two levels: aspect extraction and aspect-based classification. Several methods of aspect extraction such as nouns and noun phrases identification followed by PMI [47], balances category feature (BCF) and the other is the categorical proportional difference (MCPD, [21], sequential learning method [69] such as Hidden Markov Models (HMM, [101], [54] and Conditional Random Fields (RF, [61], frequency-based approach for emotional noun phrases [11], C-value measure-based method to retrieve multi-word aspects [158], dependency parser [159] and double propagation method [100] are discussed in the literature.

Recently, researchers have started using various evolutionary algorithms [84], semi-supervised [38], [85], [21], supervised approaches such as SVM [155], RNN [116], CNN [150], [95], [74], deep RNN [49], pre-trained word embeddings [149], LSTM [122], SLSTM [134], Memory network [136], deep recurrent belief network [22], multi-task learning network (IMN, [44], and unsupervised approaches [14] such as pLSA (Probabilistic Latent Semantic Analysis, [46], LDA (Latent Dirichlet allocation, [92], LSA based aspect-sentiment mixture model [79], joint topic sentiment model [65] for aspect extraction.

In supervised learning, attention networks are gaining popularity in the field of emotion mining, Several attention networks Attention Encoder Network (AEN, [117], relative position attention network (RPAEN, [146], Multi-grained attention network (MGAN, [32], Bi-GRU-based Position-aware Bidirectional Attention Network (PBAN, [40], recurrent attention mechanism [24], attention mechanism with GRU [43] are being used in the literature. Machine learning methods were commonly used in emotion mining but require a substantial amount of training data to achieve improved accuracy [18]. In view of the lack of large amounts of annotated data, researchers started working on ontology [98] based methods. They used a combination of hierarchical learning (HL) process and sentiment ontology tree (SOT, [140], knowledge-based along with linear SVM, or a review-based and a sentence aggregation algorithm [28], to categorize the data on the basis of emotions.

4.2. Context-Aware Emotion Mining. The digital world is becoming more informal. People use new words or phrases to express themselves. The emotional lexicons (such as NRC, WordNet, SenticNet, etc.) are not acquainted with such new words/phrases. Though, humans can interpret the meaning of such new words/phrases with the help of the surrounding context. However, in any computational model, these unseen words/phrases are simply ignored, and hence the expression of emotion may get lost. To make the computational model more robust, the need arises to make use of semantics to extract the contextual meaning.

Researchers spent lots of time exploring and selecting regular features. Several emotion-mining tools rely on emotion lexicons to support linguistic resources. These lexicons contain polarity values and some weight to emotional words. But, it is observed that emotion mining requires more evolutionary approaches that may lead to semantic-level analysis [19]. A word may change its polarity based on its context. With this idea, Gangemi et al. [37] gave importance to emotion contextualization and consider it as a major challenge in the field of emotion mining. Some of the common context-aware approaches used by researchers are Rule-based approaches [111], Sentence and discourse-based context shifters, Linguistic patterns [147], and Vector space modeling [153].

A process to segregate contextualized emotion lexicons i.e. a way to isolate sentiment terms from stable polarity terms. This process allows changing the ambiguous word's polarity as the context changes in some other textual information. With the evolution of contextual knowledge, traditional emotion lexicons are reinforced with context knowledge [73]. An iterative regression and a random walk method are used to label ConceptNet [119] with emotion values [126]. Several studies have been done to develop contextualized, cross-domain lexicons with sentiment mining and various decision-support applications [141]. It was assumed that disambiguation and contextualization can give better performance on cross-domain emotion mining using ontological lexicons. These lexicons can be used at three levels. The first level detects an ambiguous word with the help of existing labeled corpora. The second level determines the emotion score of co-occurring contextual terms and finally at the last level determines the combined polarity values for ambiguous and unambiguous terms. Experimentation and evaluation were performed on various datasets such as product reviews, hotel reviews, and movie reviews and successfully achieved better performance on each dataset. They extended the process by using the group concept from SenticNet, ConceptNet [119], Freebase [13], and DBpedia [10]. Recently, deep learning-based methods are becoming more popular as they have greatly improved the performance of emotion mining. Deep learning approaches use the structure and context of the sentence, instead of just emotion-bearing words in isolation.

Often context-based emotion mining is domain-specific and is based on the assumption that words always have the same association in that domain. For example, the use of the word unexpected: may signify positive emotion when describing the plot of an action film ("the film had an unexpected ending"), while it is not a good sign in the sense of the battery life of a laptop. Though, this may not always be valid. Consider a "leaves" description that says "The children love to play in the leaves but they do not like it when their father leaves for work." One can easily see that within the same domain, in the first case the word "leaves" communicates positive emotion, and in the second case a negative emotion. It is, therefore, reasonable to conclude that domain considerations alone are not enough to conduct context-based emotion mining.

Word embedding is becoming more popular, which is a low dimensional continuously-valued vector representation of words and is explored extensively by researchers [142], [81] The most commonly used word embeddings are CBOW, SkipGram, and GloVe [132]. Nguyen [88] proposed a semi-supervised approach to utilize an immense amount of available unlabeled data. He represented the semantic similarity of textual and user context using deep learning and latent low-dimensional space representation. He used networked characteristics also besides textual content i.e. the way users are connected via social relationships because this conveys the emotional behavior of connected users. A semantic model [138] is proposed to monitor the fluctuating emotional phases or emotions of learners while participating in MOOC's online courses. The authors used machine learning and semantic network approach in real-time to know learners' emotions about courses. Observing the change in the emotional states of a learner during the course can improve the graduation probability.

4.3. Critical analysis of Contextual Based Emotion Mining. Though lots of work has been carried out in the field of context-aware emotion mining, still it is far from the human touch. Streaming data is still a challenge, as contextual information is difficult to deal with. Nowadays, most organizations need aspect-level emotion analysis for feature details. It is observed that machine learning and deep learning with word embeddings are widely used by researchers for aspect-based emotion mining. However, accuracy and precision are still an open challenge as the existing methods/algorithms are still not able to deal with complex problems. Most of the cases are handled by emotional words. However, words or information are highly diverse, countless, and exceptional. It becomes very difficult to learn patterns via statistical methods only because there are several ways to express emotions. Lots of research, so far, mainly focus on products, movies, and restaurant reviews as well as Twitter data. In these domains, fairly good accuracy can be achieved. This is because the reviews are short and rich in emotions. However, for the other domains such as health, politics, forum discussions, commentaries, etc., the situation becomes very difficult as the emotions are either objective or factual statements in these domains.

Sometimes, the sentences are very complex due to a mixture of objective and subjective statements or sarcastic sentences. Besides this, all the social platform data is very noisy, and consists of various errors, whereas, all the analysis tools require clean data for analysis purposes. Therefore, a lot of focus is required for pre-processing tasks and methods. Some researchers incorporated the ontology method due to the nonavailability of training data, but ontology features are not robust. The ontology method also needs training data to interpret them. It is very difficult and time-consuming to manually build ontology; therefore, there is a need to automate the process to create the ontology for multiple domains. However, due to the multidisciplinary area, emotion mining takes advantage of psychology also. As per psychology, the successful evolution of online platforms is the ability to bring people together and serves very different needs of people.

Chen et al.[23] explored how opinion about any object gets affected by the user's connection with their friends, family, or relatives. They have developed a content-based sequential opinion influence framework to monitor users' behavior on social media. This framework uses an RNN to seize the opinion expressed in the sequence. The user's opinion is checked with the past information that the user acquired from his/her relation with others and his/her personal opinions. Based on the learned influence, the user's future influence prediction is determined. This valuable information can be used by companies for their future business planning.

5. Cognition-based Textual Emotion Mining. With the evolution in the field, researchers have started using cognitive features for emotion mining. As per Izard et al. [50], cognition alludes to the mental process that is impacted by emotions (desires and feelings). Human emotions (such as feelings, moods, and motivations) can impact both the way they think and the choices/decisions they make. Cognition is one of the most important events that operate and control emotions. Cognition-based emotion mining can provide us with more insights into the emotion behind words. Broadly, cognition-based emotion mining is classified into two categories: Belief based and semantic-based. They are discussed in brief in the following sections.

5.1. Belief Based Emotion Mining. Understanding mental states such as the beliefs and goals of a person enables an individual to explain behavioral patterns. According to psychologists, beliefs impact the arousal of emotions and emotions in turn influence beliefs [36], [30], Ortony et al. [91], concentrate on the cognitive elicitors of emotions [53], [51] in their hypothesis. They proposed that emotions are positive or negative reactions to their beliefs or perceptions of reality [52]. Thus, one may be glad or dissatisfied with the outcomes of an event; one can support or oppose a person's activities, and one can like or dislike the characteristics of an item. The human emotional process, according to Lazarus [62], is made up of two distinct processes: evaluation (which describes a person's connection with their environment) and coping (which suggests strategies for altering or maintaining this relationship). Both of these processes are aided by cognition. Cognition aids evaluation by constructing mental representations of how experiences relate to inherent nature or temperament such as beliefs and goals. Coping deals with the situation through suggestions and exploration of methods for changing or sustaining the individual social relationship.

In a recent study, Masland et al. [78] examined the impact of emotional reasoning on reliability assessments in people with a moderate personality disorder. The findings show that in comparison to the control group, participants with suspected moderate disorder produced more emotionally fragile evaluations and were more affected by negative experiences or beliefs. As a result, persons with moderate characteristics may be impacted differently and more precisely by unfavorable afflictions. The mental state, if it is subjected to mental illness, then the diagnosis is sometimes difficult since it necessitates detailed and in-depth psych evaluations by trained psychiatrists at an early stage [108], as well as interview sessions, lengthy questionnaires, self-reports, or evidence from friends and family. In such cases, it becomes very important to understand the underlying emotion, feelings and behavior shown by patients. Furthermore, it is highly usual for patients suffering from mental illness often avoid going to health centers to seek medical care in the early phases of their illness [161]

Hamad et al. [160] proposed the automatic depression method, which extracts depressive content using various strategies such as key phrase matching and document summarization, named entity recognition (NER), etc. on the user tweets, resulting in more granular and relevant content, which is then directed to a deep learning framework consisting of convolutional neural networks associated with attention-enhanced gated recurrent units. Vetriselvi et al. [131] provided a review of cognitive-based emotions and insights on the importance of cognitive theories especially cognitive and intuitive theory to improve emotion classification. Long et al. [72] proposed an attention-based neural network model for sentence-level sentiment classification. They trained this model with cognition-grounded eye-tracking data and built a cognition-based attention (CBA) layer for emotion mining. Further research with formal models is needed so that future systems can usefully detect emotions and accurately predict behavior.

In light of the present COVID-19 epidemic, where religion, religiosity, and emotions play crucial roles in coping, it follows that a connection between religiosity—which encompasses a variety of religious activities and beliefs—and people's emotional states and assessments of their general well-being should be obvious. Regardless of their allegiance with a particular religion, people in a wide range of geographical areas show a wide range of emotions through their religious behavior, which is inextricably linked to a formal or informal belief in God or the Divine.

5.2. Semantic / Sentic Based Emotion Mining. Emotion mining at the conceptual level while considering the common sense or semantics of the text is called sentic/semantic-based emotion mining. In 2013, Raina [[102] proposed the sentiment analyzer where emotions can be associated with a common-sense knowledge base. The author analyzed sentiments in news articles. In 2018, Ferru et al. [33] used machine learning approaches to classify a message as positive or negative towards a particular topic and on a five-point scale. They used a cognitive computing tool (i.e. IBM Watson) to extract the semantic features. Their experimental results showed that semantic features play a very important role in classification. However, the authors would like to further research semantic features extracted from other cognitive computing systems.

Irony detection is one of the challenging tasks in the field of emotion mining as it has an ambiguous interpretation. Vijay D. et al. [12] addressed the problem of irony detection. From a cognitive perspective, it is a test to think about how humans utilize irony as a specialized instrument to communicate. To detect irony, the authors constructed a Hindi-English code-mixed corpus using tweets. They used various feature vectors such as character N-gram, word N-gram, laugh words, and emoticons. In 2018, Corriga et al. [25] proposed a tool that accepted an image as information and did the classification based on the person's gender and afterward recognized their emotions.

As per Cacioppo et al. [17], it is related to a person's tendency to engage in and enjoy thinking. Das et al. [26] mentioned in the paper that it has a moderating effect on variables such as human behavior, intent to purchase, and also web surfing. Li [64] used cognitive theories for emotion mining on the linguistic feature, where sentiments are classified as target-dependent and target-independent. Zou et al. [162] proposed a strategy utilizing indirect relations in specific user structure likeness to analyze opinion. The authors experimented and verified that similar users connected via common friends have similar opinions. The authors used the sociological phenomenon of homophily to understand the connection between similar people. Their experimental results showed that indirect relation (through some common friend) has a better performance than user-direct relations to improve the accuracy of sentiment classification.

Social media has recently attracted medical natural language processing researchers to detect various medical abnormalities such as depression, anxiety, etc. Recently, many researchers explored the application of emotion mining in the medical field. Emotion mining can be used in providing healthcare assistance. Nowadays depression is one of the most prevailing issues which is often talked about on social media. Zucco et al.[163] proposed affective computing and emotion-mining methodologies to monitor depression conditions. The authors used mobile technologies to collect input data. Many researchers are working towards detecting loneliness [129], [118], depression [106], [9], [104], [105], or suicidal thoughts [75] However, the results obtained are highly data-dependent and far from cognitive theories of emotions.

5.3. Critical Analysis of Cognition-Based Emotion Mining. Very few researchers have addressed cognition-based emotion mining in recent years. Cognitive approaches talk about the human mind and behavior. Such approaches talk about how emotions are produced and what their effects are. One's belief [112] plays an important role in emotion modeling. Someone can have many emotions at the same moment, each with varying degrees of intensity. The construction of a computationally efficient model of emotion is an extremely complicated reality. For instance, "it is raining outside" may convey positive or negative emotions depending upon the person's belief. If someone believes that rain is good in pursuit of his goal, then this sentence expresses positive emotion else negative.

By reviewing and incorporating significant information, it is studied how intelligence analysis better explains challenging circumstances and boosts useful insights for effective decision-making. Intelligence analysis may assist in determining patterns and social behavior. For this kind of analysis, a high-level cognitive theory of emotions [89] can be used. Unfortunately, this area of research is still untouched. Moreover, cognition-based emotion mining can do wonders in the healthcare domain. These studies can impact a patient's life quality



Fig. 6.1: Generalized Framework of Emotion Mining from Textual Data

and health status. This study helps in knowing how emotions influence social relationships, reasoning, memory, and their function in psychological illness/disorder.

6. Summarization and Discussion. In the literature survey, researchers have applied different techniques to identify emotions from unstructured text. It is observed that all the models follow a common process flow (as shown in Fig. 6.1). At first, the unstructured data is preprocessed and cleaned to make it ready for processing. This preprocessing may involve many steps such as stop word removal, stemming, normalization, removal/replacement of slang and abbreviations, etc. Next, features are extracted and feature vectors are generated by various means such as using word embeddings, POS tagging, PMI, TFIDF, etc. Once, a feature vector is ready, it is fed to a learning model to get the output. Emotion mining is the most widely studied topic. However, emotion mining at the granularity level has been the subject of a lot of research studies. The limited number of studies on context-aware, semantic-based, and cognitive perspective is carried out because it requires knowledge of cross-domain and criticality of human nature/human perspective. There are several approaches to performing emotion mining; the machine learning approaches outperformed the traditional approaches. It is also seen that some supervised classifiers such as Support Vector Machine, Neural net, and Naïve Bayes have repeatedly been applied. The effectiveness of these classifiers is probably the reason behind this.

The same thing can be said about the used features; word stem and n-grams are frequently chosen as features. Regarding datasets, most of the researchers have used Twitter data and the IMDB dataset while some researchers built and used a new dataset. Thus, no common dataset was used for benchmarking results and evaluating experiments. In the early days, researchers analyzed texts collected from the web and focused on the word, sentence, or document level. By contrast, researchers have recently treated principally social media texts and dealt with deep learning. Very few attempts have, however, been concerned with Affective Cognitive Emotion Mining. These tasks still need much deeper investigation and research.

Although research on emotion mining started around the year 2000, studies on this issue have shown an active rise in the last few years. This is mainly due to the exponential growth of social media, online reviews, and social networking sites. The literature survey carried out in the previous sections has revealed that researchers have dealt with different emotion mining tasks: subjectivity classification, opinion classification, aspect-based emotion mining, building lexicons/resources, extracting opinion holders, belief-based emotion mining, context-aware and cognition-based emotion mining (as shown in Fig. 6.2). The excerpt of some important papers is summarized in Table 6.1. The table contains information about the dataset and methodology used. Moreover, the pros and cons of each paper are also discussed.

806



Fig. 6.2: Different Approaches for Emotion Mining over the Years

Ref	Approach	Level	Dataset	Algorithms	s Advantage	Disadvantage
[44]	Supervised	Aspect- level	Restaurant reviews, twitter reviews, and Laptop reviews	Interactive multitask learning network.	Attentional encoder network pre- vents recurrence and uses attention- based encoders for context-to-target modeling. Raised the label unrelia- bility issue. Demonstrate the effec- tiveness and light-weight of the pro- posed model.	The approach lack in calcu- lating the hid- den states.
[77]	Supervised	Character level, Word level	Hotel, Restaurant reviews	Aspect ori- entation rules	1.Annotated corpus will be freely available. 2.The corpus can be anno- tated with part-of-speech tags at the word- level which could yield better results.	1. Not gener- alized to many languages.
[115]	Semi- supervised	Context- aware	Amazon.com and IMDb.com	SVM and Naïve Bayes approach	1.Provides valuable background in- formation for advanced emotion mining. 2. Provides a platform to identify ambiguous sentiment terms, and extract context information for disambiguating	1. Lack of emphasis on scalability and throughput.
[48]	Unsupervise	Document- Level	Stanford Twitter, Sentiment (STS) and ObamaM- cCain Debate (OMD).	Semisupervi deep learn- ing, a bilinear embed- ding model (CaTER method)	 sell Provides the effectiveness of exploiting user context information for leveraging social media emotion mining using social relation graph. 2. Best accuracy among all the other methods like SVM, Distant Supervision (DS), Label Propagation (LPROP) and Logistic Regression with word embedding. 	1.Unavailability of unlabelled data for word representa- tions.
[150]	Pretraining and multi- task learning	Document- level	SemEval 2014, 2015, and 2016	CNN with gating mecha- nisms	Improved word representation in aspect-level sentiment classification.	Unable to handle sarcas- tic and irony in sentences

Table 6.1: Different Approaches for Emotion Mining over the Years

[107]	Supervised learning	Document- level	Twitter data set, political forum	Linguistic based ap- proaches, SVM	1. Bigram and trigram feature significantly outperformed unigram features for the forum data. 2. Emo- tion mining improved. 3.Sarcasm detection.	1.Domain- specific Latent Sentiments.
[72]	Cognition based	Document level/sentenc -level	IMDB, Yelp cel3, Yelp 14 and IMDB2	Cognition based attention model, LSTM	1. It works well both at a sentence and at a document level. 2. Gives higher weights to the sentiment- linked words.	1.Domain de- pendent.
[2]	Supervised	Sentence- level	Online tex- tual corpus	Emotional ontol- ogy ap- proaches	Emotional ontology improves the re- sult. Provides a platform to trans- late emotional dimensions into emo- tional categories.	1. Limited scope for speech output
[37]	Unsupervise	d Sentence- level	MPQA opinion corpus	Heuristic graph, ML, Knowl- edge representa- tion	1. The approach deals with a factual sentiment. 2. Assists in identify- ing relational features. 3. Provides improved performance. 4. Domain- independent	1. Does not perform well for noise data and sarcasm.
[8]	Semi- supervised	Sentence- level	Yahoo! Fi- nance, Rag- ing Bull.	Multi- View classifica- tion	1.Can mine a large corpus of domain-specific sentiment expres- sions	1. Does not support cross- domain.
[69]	Supervised	Sentence- level	Twitter dataset	Support Vector Machine (SVM)	1. Using related tweets along with the current tweets outperformed the results. 2. Utilized graph-based optimization techniques to improve performance.	Relations be- tween a target and any of its extended tar- gets are not explored.
[58]	Supervised/ unsuper- vised	Sentence- level	Moview & customer reviews, TREC question Dataset, Google News	Convolution Neural Network	all. Use of pre-trained vectors 2. One layer of convolution has improved the performance.	1.Hyperparamet tuning is re- quired.
[33]	Supervised cognition- based	Topic- based	Twitter and anno- tated using Crowd- Flower	Decision Trees, Linear Regression and Naive Bayes	1. Successfully classified the mes- sage on a two-point scale.	1. Dataset is unbalanced. 2.The scarcity of training data
[41]	Supervised	Word-level	Pattern from pages www.allthewe m	Point-wise mutual in-	1. Lexical patterns that are precise enough for finding emotion-bearing, affect words. 2. Re-use of the SOPM\I formula	1. Only a few emotive patterns for adjectives are used

[146]	Unsupervise	d Aspect-	Emotion	Attention-	Explored the complex multiangle	1. The size
		level	Dictionary	based LSTM	analysis of transitions in public emo- tions under the pandemic scenario. Helpful in recognizing public behav- ior and the prevention and regula- tion of pandemic situations. The is- sue of identifying polarity in the dic- tionary is strengthened effectively.	of the specific dictionaries chosen is too small.
[148]	Supervised	Character Level	Microblogs	CNN	1. Using the word vectors yield higher accuracy than those using words as feature elements. 2. Effec- tively improve the overall accuracy	Proposed model per- forms better on Chinese text but not generalized on other languages
[82]	Supervised	Word-level	EmoInt	SVM, CNN, LSTM and BiL- STM	1. Automated robust feature repre- sentation rather than manually en- gineered features. 2. Emotion di- mensionality score provides a reli- able and fine- grained analysis of text instead of just assigning the dis- crete emotion class. 3. All the data is made freely available	1. Manually identified the intensity score of basic emotions, therefore need to automate the process.
[59]	Supervised	Document- level	Tweets	RNN	1.Use of word embeddings improved the performance by reducing the high dimensional vectors.	1.Dataset size is too small to exploit deep learning Techniques. 2.Compara- bility and generalization is lacking
[113]	Rule Based	Document- level	ISEAR	Semantics rules	1.Paid special attention to Phrasal verbs.	Contextual meaning is missing. Not enough terms in the vocabulary.
[128]	Supervised	Document- level	Tweets with NRC emotion lexicon	SVM	1. Identify actionable emotion pat- terns in Tweets, with results of 84.92 percent and 88.01 percent accuracy.	1. Due to the limited number of emotional class, general- isation is not possible.
[94]	Supervised	Document- level	Tweets and ROC story Data	CNN (DL)	1. An embedding emotion model is created with the best accuracy of 73.3% for a positive emotion such as a pleasure and lowest for sadness around 36.7 %	1. Negative sentences did not perform well. 2. Nega- tion handling could improve the accuracy.

[114]	Supervised	Aspect level	Publically available data	LRA- DNN	1. Compared to the current ANN, DNN, and CNN approaches, LRA- DNN gets the highest performance with accuracy, sensitivity, and speci- ficity at rates of 94.77%, 92.23%, and 95.91%, respectively 2. It effec-	1. Execution time may get high while assigning rank to the given feature.
					tively lowers categorization and mis- prediction errors.	
[130]	Unsupervise	-	Product re-	LDA	1. Proposed a filtered and strong	1. The use of
	Approach	Level	views	Model	semantic approach, successfully ex-	regular expres-
					tracted relevant aspects from prod-	sion makes
					uct reviews.	the approach
						little complex.
[63]	Supervised	Sentence	Food Com-	Bidirectiona	1. This sort of approach exam-	The approach
		Level	ments	long-term	ines, interprets, infers, and analyses	is not com-
				and short-	the emotional remark content before	pared and
				term	coming to the appropriate conclu-	generalized.
				memory	sion.	
				network		
				(BiLSTM)		

7. Challenges in Text-based Emotion Mining. Text is a challenging medium for analyzing hidden emotions. Due to the high complexity and the raw data available online, the area is still lacking. This section summarizes the challenges which still need the attention of researchers. The same has been depicted pictorially (Fig. 7.1). The open challenges are:

- 1. Domain dependence: Existing approaches mainly focus on the opinionated text, where the sentiments and emotions are explicitly expressed. Syntactical approaches have been used as the text is domain-dependent. Therefore, many language patterns are missed. Therefore, there is a need to focus on the deep analysis of the semantics of the sentences containing implicit emotions. At the same time, the emotion mining model should be domain-independent.
- 2. Dynamic effective model: To adjust the system to the social platform's user affectivity, the system should identify the user's emotions. The system needs to know when the user is negative about their service/ topic/ product/policy etc. The services or products must be customized as per the user's opinion regarding the product/ services. The affective history of the author should be maintained in the affective user model. It helps to respond appropriately to the user's emotions. Therefore, we need a dynamic affective model. The model should consider the changes in the emotional states of the user, to improve the accuracy significantly. Developing a dynamic affective model is very challenging because of the complex human cognitive states.
- 3. Feature engineering: People convey their emotions in complex ways like sarcasm, irony, etc. Such sentences do not specifically communicate their meaning but implicit meaning can mislead emotion mining. The only way to understand this issue is through context. Though a lot of research has been conducted in context-based emotion mining, there is a need to focus more on feature engineering as little attention has been paid to it so far. Feature engineering creates new features in a model. These features help to better understand the context and emotions of the users. Engineered features can help to boost accuracy and get an overall sense of what the social platform's user is trying to convey.
- 4. Context extraction: Truthful words and expressions inferring emotions have scarcely been considered, but they are exceptionally useful for numerous domains. The contextual information of emotional words stays to be highly challenging indeed with so much research.
- 5. Unknown/unseen words: The large vocabulary of words is available widely. However, it is a very tedious task to manually annotate these huge numbers of features. Moreover, unseen words in the text also cause problems. Therefore, it is very important and challenging to establish domain-independent affective words that can be used in language transfer.
- 6. Labeled data: Accessing the labeled data to train the model is very difficult. For the best long-term results, it is suggested to train our models. Therefore, need to find out ways to access data quickly.



Fig. 7.1: Challenges in Textual Emotion Mining

- 7. Different sentences: Nevertheless, the question type of sentences, which do not depict any opinion about any particular entity such as "Are you serious?" "Have you gone mad?" also conveys significant emotions and should be handled.
- 8. Belief incorporation: For the same event or product, two people can express different emotions or the same person can express different emotions at different instances of time. This is due to the difference in beliefs and expectations. To date, no textual emotion mining model considers these parameters

8. Opportunities. As discussed, the practice of collecting and understanding human emotions from diverse data sources, such as text, audio, and video, is known as "emotion mining". However, we have limited our study to text data. While the field of emotion mining has advanced significantly in recent years, there are still a lot of untapped or understudied areas. Here are a few innovative prospective research areas:

- 1. Emotional State Prediction: Construct predictive models capable of anticipating fluctuations in emotional states by leveraging historical data. These models have versatile applications spanning mental health monitoring, customer service optimization, and educational contexts.
- 2. Emotion Mining for Mental Health: Use emotion mining methods to aid in diagnosing and keeping track of mental health problems. This might mean looking at what people write or how they talk (speech data) on social media to find signs of depression, anxiety, or other mental health concerns.
- 3. Human-Robot Emotion Interaction: Delve into the application of emotion mining within human-robot interaction scenarios, empowering robots to gain a deeper understanding of human emotions and respond more effectively. This can significantly enhance their performance in tasks like detecting an early risk of emotion disorders, caregiving, and customer service.
- 4. Emotion Mining's Biases and Ethical Considerations: Study the moral questions raised by emotion mining, like how it affects people's privacy, whether they've given permission, and whether the algorithms that recognize emotions are treating everyone fairly. Create rules and guidelines for doing emotion mining that are ethical and fair.
- 5. Understanding Emotions in Legal and Forensic Contexts: Explore how emotion mining can be applied in legal situations and forensic investigations. This involves examining voice recordings or written statements to identify signs of deception or emotional distress.

9. Applications. Emotion mining has made it easier to estimate others' emotions, feelings, and psychology. Based on the result generated through emotion mining, one can always adjust/control the present situation and manage customer needs in a better way. It allows for staying dynamic throughout. Due to its various advantages, it has been connected to numerous application zones. It has been used in various fields such as the medical healthcare domain, sports, education, the financial sector, online social media, politics, hospitality, tourism, and many more. Some of the emerging application areas are briefed below:

1. Medical health-care domain: Using textual emotion mining, it is possible to assess the psychological health of a person with the help of his online posts [145] and avoid suicide to some extent. Moreover, it can be used to analyze the reaction of people to some disease outbreaks [57] and come up with some feasible solutions.

- 2. Sports domain: Emotion mining can be used for the spatial relationship between crime events and Twitter activity in the context of any popular game [109] Also, it can be used to assess the passionate reactions of fans [156].
- 3. Education domain: It can be used by the instructor to monitor the student's overall interest over time in the course [90] which can further help to improve the teaching and learning system [103]
- 4. Politics: Emotion mining can be used to predict the election results [107] and the reaction of citizens to any new policy.
- 5. Hospitality and Tourism: To improve the hospitality sector, customer reviews play an important role. Emotion mining can be used to predict customers' orientation based on reviews they posted on social media [77], [96].

10. Conclusion and Future Directions. The process of identifying an individual's mental state/ emotions/ opinions/ sentiments is an ongoing field of research. Most organizations are availing of it and incorporating it into their processes. A plethora of surveys can be found in the relevant literature either focusing on conventional approaches or machine learning approaches. Online social networking platforms have expanded exponentially over the past few years and consequently generated an enormous amount of data. Now, the attention is rapidly escalating to the problem of how to analyze the available data. This paper provides a systematic survey of the state-of-the-art approaches for emotion mining for textual data. Besides polarity identification, emotion mining also about knows the interpersonal relationships in the context of social networks, which affect information flows such as identifying behavioral features that may contain a strong emotional signal [15]. Accordingly, emotion mining needs to do deeper semantic analysis as social online platforms are very noisy.

In the future, research can be conducted in the direction of analyzing or recognizing the emotions hidden in the text understandably to recognize various aspects of human nature and behavior. This research can provide very predictive influences on the public attitude towards various issues and can also explain the state of human mental well-being. Further, information repositories of texts such as online chats, blogs, forums, etc., which are the source of opinioned or emotional content, are expected to give more emphasis on sincerity, spontaneity, and effectuality by combining the models of human cognitive capabilities that may include emotion mining.

Moreover, a human's mental state approach, more specifically the belief-desire-intention (BDI) model, can be used to know the human's intentions or behavior, which may be helpful to predict the human action pattern more accurately and improve emotion prediction. The main goal of using this approach is to capture emotions from the cognition viewpoint. Moreover, in today's scenario, only domain-dependent keywords are not sufficient, we need to explore a commonsense knowledge base that extends the cognitive and affective information associated with the social platform's information. Future emotion mining strategies need broader common-sense approaches which should be driven by the process of human cognition so that the effects are more realistic, reliable, and related to human psychology. We hope that this survey serves as a starting point for much more to come.

REFERENCES

- [1] http://nlp.uned.es/~lplaza/papers/CONLL_2010.pdf. Accessed: 2023-3-22.
- M. ALMASHRAEE, D. M. DÍAZ, AND A. PASCHKE, Emotion level sentiment analysis: The affective opinion evaluation, EMSA-RMed@ESWC, (2016).
- [3] A. ANDREEVSKAIA AND S. BERGLER, Semantic tag extraction from WordNet glosses. http://www.lrec-conf.org/proceedings/lrec2006/pdf/566_pdf.pdf. Accessed: 2023-3-22.
- [4] A. H. AWADALLAH, V. QAZVINIAN, AND D. R. RADEV, What's with the attitude? identifying sentences with attitude in online discussions, in EMNLP 2010 - Conference on Empirical Methods in Natural Language Processing, Proceedings of the Conference, 2010, pp. 1245–1255.
- [5] F. BARBIERI AND H. SAGGION, Modelling irony in twitter, in Proceedings of the Student Research Workshop at the 14th Conference of the European Chapter of the Association for Computational Linguistics, Stroudsburg, PA, USA, 2014, Association for Computational Linguistics.
- [6] K. C. BATHINA, M. TEN THIJ, L. LORENZO-LUACES, L. A. RUTTER, AND J. BOLLEN, Individuals with depression express more distorted thinking on social media, Nat. Hum. Behav., 5 (2021), pp. 458–466.
- [7] G. BELLSTAM, S. BHAGAT, AND J. A. COOKSON, A text-based analysis of corporate innovation, Manage. Sci., 67 (2021), pp. 4004–4031.
- [8] Z. BEN-AMI, R. FELDMAN, AND B. ROSENFELD, Using multi-view learning to improve detection of investor sentiments on twitter, Comput. Sist., 18 (2014).
- [9] S. S. BISHT, H. SHANDILYA, V. GUPTA, S. AGRAWAL, AND S. JAIN, Perceiving the level of depression from web text, in Artificial Intelligence, Machine Learning, and Mental Health in Pandemics, Elsevier, 2022, pp. 277–298.

812

- [10] C. BIZER, J. LEHMANN, G. KOBILAROV, S. AUER, C. BECKER, R. CYGANIAK, AND S. HELLMANN, DBpedia a crystallization point for the web of data, Web Semant., 7 (2009), pp. 154–165.
- [11] S. BLAIR-GOLDENSOHN, T. NEYLON, K. HANNAN, G. A. REIS, R. MCDONALD, AND J. REYNAR, Building a sentiment summarizer for local service reviews, NLPIX2008, (2008).
- [12] A. BOHRA, D. VIJAY, V. SINGH, S. S. AKHTAR, AND M. SHRIVASTAVA, A dataset of Hindi-English code-mixed social media text for hate speech detection, in Proceedings of the Second Workshop on Computational Modeling of People's Opinions, Personality, and Emotions in Social Media, New Orleans, Louisiana, USA, June 2018, Association for Computational Linguistics, pp. 36–41.
- [13] K. BOLLACKER, C. EVANS, P. PARITOSH, T. STURGE, AND J. TAYLOR, Freebase, in Proceedings of the 2008 ACM SIGMOD international conference on Management of data, New York, NY, USA, June 2008, ACM.
- [14] S. BRODY AND N. ELHADAD, An Unsupervised Aspect-Sentiment Model for Online Reviews, 2010.
- [15] J. J. BROWN AND P. H. REINGEN, Social ties and word-of-mouth referral behavior, J. Consum. Res., 14 (1987), p. 350.
- [16] P. BURNAP, M. L. WILLIAMS, L. SLOAN, O. RANA, W. HOUSLEY, A. EDWARDS, V. KNIGHT, R. PROCTER, AND A. VOSS, Tweeting the terror: modelling the social media reaction to the woolwich terrorist attack, Soc. Netw. Anal. Min., 4 (2014).
- [17] J. T. CACIOPPO AND R. E. PETTY, The need for cognition, J. Pers. Soc. Psychol., 42 (1982), pp. 116–131.
- [18] E. CAMBRIA, Affective computing and sentiment analysis, IEEE Intell. Syst., 31 (2016), pp. 102–107.
- [19] E. CAMBRIA, N. HOWARD, J. HSU, AND A. HUSSAIN, Sentic blending: Scalable multimodal fusion for the continuous interpretation of semantics and sentics, in 2013 IEEE Symposium on Computational Intelligence for Human-like Intelligence (CIHLI, IEEE, Apr. 2013.
- [20] E. CAMBRIA, R. SPEER, C. HAVASI, AND A. HUSSAIN, SenticNet: A publicly available semantic resource for opinion mining. https://www.aaai.org/ocs/index.php/FSS/FSS10/paper/download/2216/2617.pdf. Accessed: 2023-3-22.
- [21] J.-R. CHANG, H.-Y. LIANG, L.-S. CHEN, AND C.-W. CHANG, Novel feature selection approaches for improving the performance of sentiment classification, J. Ambient Intell. Humaniz. Comput., (2020).
- [22] I. CHATURVEDI, Y.-S. ONG, I. W. TSANG, R. E. WELSCH, AND E. CAMBRIA, Learning word dependencies in text by means of a deep recurrent belief network, Knowl. Based Syst., 108 (2016), pp. 144–154.
- [23] C. CHEN, Z. WANG, AND W. LI, Tracking dynamics of opinion behaviors with a content-based sequential opinion influence model, IEEE Trans. Affect. Comput., 11 (2020), pp. 627–639.
- [24] P. CHEN, Z. SUN, L. BING, AND W. YANG, Recurrent attention network on memory for aspect sentiment analysis, in Proceedings of the 2017 Conference on Empirical Methods in Natural Language Processing, Stroudsburg, PA, USA, 2017, Association for Computational Linguistics.
- [25] A. CORRIGA, S. CUSIMANO, F. M. MALLOCI, L. MARCHESI, AND D. R. RECUPERO, Leveraging cognitive computing for gender and emotion detection., in EMSASW@ ESWC, 2018, pp. 47–56.
- [26] S. DAS, R. ECHAMBADI, M. MCCARDLE, AND M. LUCKETT, Mark. Lett., 14 (2003), pp. 185–202.
- [27] K. DAVE, S. LAWRENCE, AND D. M. PENNOCK, *Mining the peanut gallery*, in Proceedings of the twelfth international conference on World Wide Web - WWW '03, New York, New York, USA, 2003, ACM Press.
- [28] S. DE KOK, L. PUNT, R. VAN DEN PUTTELAAR, K. RANTA, K. SCHOUTEN, AND F. FRASINCAR, Review-level aspect-based sentiment analysis using an ontology, in Proceedings of the 33rd Annual ACM Symposium on Applied Computing, New York, NY, USA, Apr. 2018, ACM.
- [29] K. DENECKE AND Y. DENG, Sentiment analysis in medical settings: New opportunities and challenges, Artif. Intell. Med., 64 (2015), pp. 17–27.
- [30] M. J. DOHERTY, Theory of mind: How children understand others' thoughts and feelings, Psychology Press, Jan. 2008.
- [31] A. ESULI AND F. SEBASTIANI, SENTIWORDNET: A Publicly Available Lexical Resource for Opinion Mining. International Conference on Language Resources and Evaluation, 2006.
- [32] F. FAN, Y. FENG, AND D. ZHAO, Multi-grained attention network for aspect-level sentiment classification, in Proceedings of the 2018 Conference on Empirical Methods in Natural Language Processing, Stroudsburg, PA, USA, 2018, Association for Computational Linguistics.
- [33] D. S. FERRU, F. IBBA, AND D. R. RECUPERO, Supervised topic-based message polarity classification using cognitive computing, CEUR Workshop Proceedings, 2111 (2018), pp. 11–21.
- [34] A. FESSEHA, S. XIONG, E. D. EMIRU, M. DIALLO, AND A. DAHOU, Text classification based on convolutional neural networks and word embedding for low-resource languages: Tigrinya, Information (Basel), 12 (2021), p. 52.
- [35] E. FORSLID AND N. WIKÉN, Automatic irony- and sarcasm detection in Social media, 2015.
- [36] N. H. FRIJDA AND B. MESQUITA, Mesquita 1998 The analysis of emotions Dimensions of variation," What develops in emotional development Emotions personality and psychotherapy, 1998.
- [37] A. GANGEMI, V. PRESUTTI, AND D. REFORGIATO RECUPERO, Frame-based detection of opinion holders and topics: A model and a tool, IEEE Comput. Intell. Mag., 9 (2014), pp. 20–30.
- [38] R. GHANI, K. PROBST, Y. LIU, M. KREMA, AND A. FANO, Text mining for product attribute extraction, SIGKDD Explor., 8 (2006), pp. 41–48.
- [39] S. L. GORDON, The sociology of sentiments and emotion, in Social Psychology, Routledge, Sept. 2017, pp. 562–592.
- [40] S. GU, L. ZHANG, Y. HOU, AND Y. SONG, A position-aware bidirectional attention network for aspect-level sentiment analysis, in Proc. 27th Int. Conf. Comput. Linguist, 2018.

- [41] E. GUTIERREZ, W. KARWOWSKI, K. FIOK, M. R. DAVAHLI, T. LICIAGA, AND T. AHRAM, Analysis of human behavior by mining textual data: Current research topics and analytical techniques, Symmetry (Basel), 13 (2021), p. 1276.
- [42] V. HATZIVASSILOGLOU AND J. M. WIEBE, Effects of adjective orientation and gradability on sentence subjectivity, in Proceedings of the 18th conference on Computational linguistics -, Morristown, NJ, USA, 2000, Association for Computational Linguistics.
- [43] R. HE, W. S. LEE, H. T. NG, AND D. DAHLMEIER, Effective attention modeling for Aspect-Level sentiment classification, in Proc. 27th Int. Conf. Comput. Linguist, 2018.
- [44] R. HE, W. S. LEE, H. T. NG, AND D. DAHLMEIER, An interactive multi-task learning network for end-to-end aspect-based sentiment analysis, in Proceedings of the 57th Annual Meeting of the Association for Computational Linguistics, Stroudsburg, PA, USA, 2019, Association for Computational Linguistics.
- [45] I. HERNÁNDEZ-FARÍAS, J.-M. BENEDÍ, AND P. ROSSO, Applying basic features from sentiment analysis for automatic irony detection, in Pattern Recognition and Image Analysis, Lecture notes in computer science, Springer International Publishing, Cham, 2015, pp. 337–344.
- [46] T. HOFMANN, Probabilistic latent semantic indexing, SIGIR Forum, 51 (2017), pp. 211–218.
- [47] M. HU AND B. LIU, *Mining and summarizing customer reviews*, in Proceedings of the tenth ACM SIGKDD international conference on Knowledge discovery and data mining, New York, NY, USA, Aug. 2004, ACM.
- [48] X. HU, J. TANG, H. GAO, AND H. LIU, Unsupervised sentiment analysis with emotional signals, in Proceedings of the 22nd international conference on World Wide Web, New York, NY, USA, May 2013, ACM.
- [49] O. IRSOY AND C. CARDIE, Deep recursive neural networks for compositionality in language, 2014.
- [50] C. E. IZARD, Basic emotions, relations among emotions, and emotion-cognition relations, Psychol. Rev., 99 (1992), pp. 561–565.
- [51] S. JAIN AND K. ASAWA, Emet: emotion elicitation and emotion transition model, in Information Systems Design and Intelligent Applications: Proceedings of Second International Conference INDIA 2015, Volume 1, Springer, 2015, pp. 209–217.
- [52] ——, Programming an expressive autonomous agent, Expert Systems with Applications, 43 (2016), pp. 131–141.
- [53] ——, Modeling of emotion elicitation conditions for a cognitive-emotive architecture, Cogn. Syst. Res., 55 (2019), pp. 60–76.
- [54] W. JIN AND H. H. HO, A novel lexicalized HMM-based learning framework for web opinion mining, in Proceedings of the 26th Annual International Conference on Machine Learning, New York, NY, USA, June 2009, ACM.
- [55] R. KAMAL, M. A. SHAH, C. MAPLE, M. MASOOD, A. WAHID, AND A. MEHMOOD, Emotion classification and crowd source sensing; a lexicon based approach, IEEE Access, 7 (2019), pp. 27124–27134.
- [56] A. KHADJEH NASSIRTOUSSI, S. AGHABOZORGI, T. YING WAH, AND D. C. L. NGO, Text mining for market prediction: A systematic review, Expert Syst. Appl., 41 (2014), pp. 7653–7670.
- [57] K. G. KIM, Book review: Deep learning, Healthc. Inform. Res., 22 (2016), p. 351.
- [58] S.-M. KIM AND E. HOVY, Determining the sentiment of opinions, in Proceedings of the 20th international conference on Computational Linguistics - COLING '04, Morristown, NJ, USA, 2004, Association for Computational Linguistics.
- [59] B. KRATZWALD, S. ILI, M. KRAUS, S. FEUERRIEGEL, AND H. PRENDINGER, Deep learning for affective computing: Text-based emotion recognition in decision support, Decis. Support Syst., 115 (2018), pp. 24–35.
- [60] A. LABIDI, F. OUAMANI, AND N. B. BEN SAOUD, An ontology based text approach for culture aware emotion mining: A modele plugin, in 2021 International Conference of Women in Data Science at Taif University (WiDSTaif), IEEE, Mar. 2021.
- [61] J. D. LAFFERTY, A. MCCALLUM, AND F. C. N. PEREIRA, Conditional Random Fields: Probabilistic Models for Segmenting and Labeling Sequence Data, 2001.
- [62] R. S. LAZARUS, Emotion and Adaptation, Oxford University Press, New York, NY, Apr. 1994.
- [63] Y. LI, X. J. YANG, Y. LIU, R. J. LIU, AND Q. Y. JIN, A new emotion analysis fusion and complementary model based on online food reviews, Computers & Electrical Engineering, 3 (2022), pp. 1–12.
- [64] Z. LI, H. XIE, G. CHENG, AND Q. LI, Word-level emotion distribution with two schemas for short text emotion classification, Knowl. Based Syst., 227 (2021), p. 107163.
- [65] C. LIN AND Y. HE, Joint sentiment/topic model for sentiment analysis, in Proceedings of the 18th ACM conference on Information and knowledge management, New York, NY, USA, Nov. 2009, ACM.
- [66] M. LING, Q. CHEN, Q. SUN, AND Y. JIA, Hybrid neural network for sina weibo sentiment analysis, IEEE Trans. Comput. Soc. Syst., 7 (2020), pp. 983–990.
- [67] B. LIU, Sentiment analysis and opinion mining, Synth. Lect. Hum. Lang. Technol., 5 (2012), pp. 1–167.
- [68] B. LIU, Sentiment Analysis: Mining Opinions, Sentiments, and Emotions (Studies in Natural Language Processing), Cambridge University Press, 2020.
- [69] B. LIU, M. HU, AND J. CHENG, Opinion observer, in Proceedings of the 14th international conference on World Wide Web
 WWW '05, New York, New York, USA, 2005, ACM Press.
- [70] B. LIU AND L. ZHANG, A survey of opinion mining and sentiment analysis, in Mining Text Data, Springer US, Boston, MA, 2012, pp. 415–463.
- [71] P. LIU, X. QIU, AND X. HUANG, Recurrent neural network for text classification with multi-task learning, (2016).
- [72] Y. LONG, L. QIN, R. XIANG, M. LI, AND C.-R. HUANG, A cognition based attention model for sentiment analysis, in Proceedings of the 2017 Conference on Empirical Methods in Natural Language Processing, Stroudsburg, PA, USA, 2017, Association for Computational Linguistics.

- [73] Y. LU, M. CASTELLANOS, U. DAYAL, AND C. ZHAI, Automatic construction of a context-aware sentiment lexicon, in Proceedings of the 20th international conference on World wide web, New York, NY, USA, Mar. 2011, ACM.
- [74] Z. MA, Improved convolutional neural network for biomedical text classification, J. Phys. Conf. Ser., 1883 (2021), p. 012080.
- [75] M. MACALLI, M. NAVARRO, M. ORRI, M. TOURNIER, R. THIÉBAUT, S. M. CÔTÉ, AND C. TZOURIO, A machine learning approach for predicting suicidal thoughts and behaviours among college students, Sci. Rep., 11 (2021).
- [76] X. MAO, S. CHANG, J. SHI, F. LI, AND R. SHI, Sentiment-aware word embedding for emotion classification, Appl. Sci. (Basel), 9 (2019), p. 1334.
- [77] E. MARRESE-TAYLOR, J. D. VELÁSQUEZ, F. BRAVO-MARQUEZ, AND Y. MATSUO, Identifying customer preferences about tourism products using an aspect-based opinion mining approach, Procedia Comput. Sci., 22 (2013), pp. 182–191.
- [78] S. R. MASLAND AND J. M. HOOLEY, When trust does not come easily: Negative emotional information unduly influences trustworthiness appraisals for individuals with borderline personality features, J. Pers. Disord., 34 (2020), pp. 394–409.
- [79] Q. MEI, X. LING, M. WONDRA, H. SU, AND C. ZHAI, *Topic sentiment mixture*, in Proceedings of the 16th international conference on World Wide Web, New York, NY, USA, May 2007, ACM.
- [80] A. B. MERIEM, L. HLAOUA, AND L. B. ROMDHANE, A fuzzy approach for sarcasm detection in social networks, Procedia Comput. Sci., 192 (2021), pp. 602–611.
- [81] T. MIKOLOV, I. SUTSKEVER, K. CHEN, G. CORRADO, AND J. DEAN, Distributed representations of words and phrases and their compositionality, (2013).
- [82] S. MOHAMMAD, F. BRAVO-MARQUEZ, M. SALAMEH, AND S. KIRITCHENKO, SemEval-2018 task 1: Affect in tweets, in Proceedings of The 12th International Workshop on Semantic Evaluation, Stroudsburg, PA, USA, 2018, Association for Computational Linguistics.
- [83] S. M. MOHAMMAD, Sentiment analysis, in Emotion Measurement, Elsevier, 2016, pp. 201–237.
- [84] M. E. MOWLAEI, M. SANIEE ABADEH, AND H. KESHAVARZ, Aspect-based sentiment analysis using adaptive aspect-based lexicons, Expert Syst. Appl., 148 (2020), p. 113234.
- [85] A. MUKHERJEE AND B. LIU, Aspect Extraction through Semi-Supervised Modeling, 2012.
- [86] S. MUKUND AND R. SRIHARI, A vector space model for subjectivity classification in Urdu aided by co-training, in Coling 2010: Posters, Beijing, China, Aug. 2010, Coling 2010 Organizing Committee, pp. 860–868.
- [87] M. MUNEZERO, C. S. MONTERO, E. SUTINEN, AND J. PAJUNEN, Are they different? affect, feeling, emotion, sentiment, and opinion detection in text, IEEE Trans. Affect. Comput., 5 (2014), pp. 101–111.
- [88] L. M. NGUYEN, Context-aware text representation for social relation aided sentiment analysis, in Proceedings of the 25th International Conference Companion on World Wide Web - WWW '16 Companion, New York, New York, USA, 2016, ACM Press.
- [89] K. OATLEY AND P. N. JOHNSON-LAIRD, Cognitive approaches to emotions, Trends Cogn. Sci., 18 (2014), pp. 134–140.
- [90] A. ORTIGOSA, J. M. MARTÍN, AND R. M. CARRO, Sentiment analysis in facebook and its application to e-learning, Comput. Human Behav., 31 (2014), pp. 527–541.
- [91] A. ORTONY, G. L. CLORE, AND A. COLLINS, The Cognitive Structure of Emotions / Cognition /, Cambridge University Press, 1988.
- [92] A. OSMANI, J. B. MOHASEFI, AND F. S. GHAREHCHOPOGH, Enriched latent dirichlet allocation for sentiment analysis, Expert Syst., 37 (2020).
- [93] G. PANG, K. LU, X. ZHU, J. HE, Z. MO, Z. PENG, AND B. PU, Aspect-level sentiment analysis approach via BERT and aspect feature location model, Wirel. Commun. Mob. Comput., 2021 (2021), pp. 1–13.
- [94] S.-H. PARK, B.-C. BAE, AND Y.-G. CHEONG, Emotion recognition from text stories using an emotion embedding model, in 2020 IEEE International Conference on Big Data and Smart Computing (BigComp, IEEE, Feb. 2020.
- [95] M. A. PARWEZ, M. ABULAISH, AND JAHIRUDDIN, Multi-label classification of microblogging texts using convolution neural network, IEEE Access, 7 (2019), pp. 68678–68691.
- [96] K. PHILANDER AND Y. ZHONG, Twitter sentiment analysis: Capturing sentiment from integrated resort tweets, Int. J. Hosp. Manag., 55 (2016), pp. 16–24.
- [97] R. W. PICARD, Affective computing: challenges, Int. J. Hum. Comput. Stud., 59 (2003), pp. 55-64.
- [98] J. POLPINIJ AND A. K. GHOSE, An ontology-based sentiment classification methodology for online consumer reviews, in 2008 IEEE/WIC/ACM International Conference on Web Intelligence and Intelligent Agent Technology, IEEE, Dec. 2008.
- [99] S. PORIA, A. GELBUKH, E. CAMBRIA, P. YANG, A. HUSSAIN, AND T. DURRANI, Merging SenticNet and WordNet-Affect emotion lists for sentiment analysis, in 2012 IEEE 11th International Conference on Signal Processing, IEEE, Oct. 2012.
- [100] G. QIU, B. LIU, J. BU, AND C. CHEN, Opinion word expansion and target extraction through double propagation, Comput. Linguist. Assoc. Comput. Linguist., 37 (2011), pp. 9–27.
- [101] L. R. RABINER, A tutorial on hidden markov models and selected applications in speech recognition, Proc. IEEE Inst. Electr. Electron. Eng., 77 (1989), pp. 257–286.
- [102] P. RAINA, Sentiment analysis in news articles using sentic computing, in 2013 IEEE 13th International Conference on Data Mining Workshops, IEEE, Dec. 2013.
- [103] S. RANI AND P. KUMAR, A sentiment analysis system to improve teaching and learning, Computer (Long Beach Calif.), 50 (2017), pp. 36–43.

- [104] T. RAWAT AND S. JAIN, A dimensional representation of depressive text, in Data Analytics and Management: Proceedings of ICDAM, Springer, 2021, pp. 175–187.
- [105] —, Depression detection: approaches, challenges and future directions, in Artificial Intelligence, Machine Learning, and Mental Health in Pandemics, Elsevier, 2022, pp. 209–234.
- [106] —, DPre: Effective preprocessing techniques for social media depressive text, Intell. Decis. Technol., 16 (2022), pp. 475–485.
- [107] M. RINGSQUANDL AND D. PETKOVIC, Analyzing political sentiment on twitter, AAAI Spring Symp. Tech. Rep, (2013).
- [108] E. A. RÍSSOLA, M. ALIANNEJADI, AND F. CRESTANI, Beyond modelling: Understanding mental disorders in online social media, in Lecture Notes in Computer Science, Lecture notes in computer science, Springer International Publishing, Cham, 2020, pp. 296–310.
- [109] A. RISTEA, J. KURLAND, B. RESCH, M. LEITNER, AND C. LANGFORD, Estimating the spatial distribution of crime events around a football stadium from georeferenced tweets, ISPRS Int. J. Geoinf., 7 (2018), p. 43.
- [110] S. SABRA, K. MAHMOOD MALIK, AND M. ALOBAIDI, Prediction of venous thromboembolism using semantic and sentiment analyses of clinical narratives, Comput. Biol. Med., 94 (2018), pp. 1–10.
- [111] I. H. SARKER AND A. S. M. KAYES, ABC-RuleMiner: User behavioral rule-based machine learning method for context-aware intelligent services, J. Netw. Comput. Appl., 168 (2020), p. 102762.
- [112] F. SCHWEITZER AND D. GARCIA, An agent-based model of collective emotions in online communities, Eur. Phys. J. B, 77 (2010), pp. 533-545.
- [113] D. SEAL, U. K. ROY, AND R. BASAK, Sentence-level emotion detection from text based on semantic rules, in Information and Communication Technology for Sustainable Development, Advances in intelligent systems and computing, Springer Singapore, Singapore, 2020, pp. 423–430.
- [114] N. SHELKE, An efficient way of text-based emotion analysis from social media using LRA- DNN", Neuroscience Informatics, (2022).
- [115] C. SINDHU, D. RAJKAKATI, AND C. SHELUKAR, Context-based sentiment analysis on amazon product customer feedback data, in Artificial Intelligence Techniques for Advanced Computing Applications, Lecture notes in networks and systems, Springer Singapore, Singapore, 2021, pp. 515–527.
- [116] R. Socher and C. D. Manning, Deep Learning for NLP, 2013.
- [117] Y. SONG, J. WANG, T. JIANG, Z. LIU, AND Y. RAO, Targeted sentiment classification with attentional encoder network, in Artificial Neural Networks and Machine Learning – ICANN 2019: Text and Time Series, Lecture notes in computer science, Springer International Publishing, Cham, 2019, pp. 93–103.
- [118] R. SOOD, H. VARMA, K. PANDEY, S. JAIN, D. SRIRAM, AND A. S. GUGLANI, Predicting loneliness from social media text using machine learning techniques, in Artificial Intelligence, Machine Learning, and Mental Health in Pandemics, Elsevier, 2022, pp. 259–275.
- [119] R. SPEER AND C. HAVASI, ConceptNet 5: A large semantic network for relational knowledge, in The People's Web Meets NLP, Springer Berlin Heidelberg, Berlin, Heidelberg, 2013, pp. 161–176.
- [120] J. STEINBERGER AND P. LENKOVA, Multilingual Entity-Centered Sentiment Analysis Evaluated by Parallel Corpora, 2011.
- [121] J. E. STETS, *Emotions and sentiments*, in Handbook of Social Psychology, Springer US, 2006, pp. 309–335.
- [122] D. TANG, B. QIN, X. FENG, AND T. LIU, Effective LSTMs for Target-Dependent sentiment classification, (2015).
- [123] D. TANG, B. QIN, AND T. LIU, Learning semantic representations of users and products for document level sentiment classification, in Proceedings of the 53rd Annual Meeting of the Association for Computational Linguistics and the 7th International Joint Conference on Natural Language Processing (Volume 1: Long Papers), Stroudsburg, PA, USA, 2015, Association for Computational Linguistics.
- [124] P. A. THOITS, Emotional deviance: Research agendas, in Research agendas in the sociology of emotions, T. D. Kemper, ed., 1990, pp. 180–203.
- [125] R. M. TONG, An operational system for detecting and tracking opinions in On-Line discussion, in Proceedings of SIGIR Workshop on Operational Text Classification, 2001.
- [126] A. C.-R. TSAI, C.-E. WU, R. T.-H. TSAI, AND J. Y.-J. HSU, Building a concept-level sentiment dictionary based on commonsense knowledge, IEEE Intell. Syst., 28 (2013), pp. 22–30.
- [127] O. TSUR, D. DAVIDOV, AND A. RAPPOPORT, ICWSM a great catchy name: Semi-supervised recognition of sarcastic sentences in online product reviews, Proceedings of the International AAAI Conference on Web and Social Media, 4 (2010), pp. 162–169.
- [128] A. TZACHEVA, J. RANGANATHAN, AND S. Y. MYLAVARAPU, Actionable pattern discovery for tweet emotions, in Advances in Intelligent Systems and Computing, Advances in intelligent systems and computing, Springer International Publishing, Cham, 2020, pp. 46–57.
- [129] T. G. VAN TILBURG, S. STEINMETZ, E. STOLTE, H. VAN DER ROEST, AND D. H. DE VRIES, Loneliness and mental health during the COVID-19 pandemic: A study among dutch older adults, J. Gerontol. B Psychol. Sci. Soc. Sci., 76 (2021), pp. e249–e255.
- [130] M. VENUGOPALAN AND D. GUPTA, An enhanced guided lda model augmented with bert based zemantic strength for aspect term extraction in sentiment analysis". Knowledge-Based Systems 246.

- [131] T. VETRISELVI AND A. VADIVEL, Sentiment and emotion prediction through cognition: A review, in Progress in Systems Engineering, Advances in intelligent systems and computing, Springer International Publishing, 2015, pp. 719–722.
- [132] T. VRBANEC AND A. MEŠTROVIĆ, Corpus-based paraphrase detection experiments and review, Information (Basel), 11 (2020), p. 241.
- [133] R. WADAWADAGI AND V. PAGI, Fine-grained sentiment rating of online reviews with deep-RNN, in Advances in Intelligent Systems and Computing, Advances in intelligent systems and computing, Springer Singapore, Singapore, 2021, pp. 687–700.
- [134] B. WANG AND W. LU, Learning latent opinions for aspect-level sentiment classification, Proc. Conf. AAAI Artif. Intell., 32 (2018).
- [135] L. WANG AND R. XIA, Sentiment lexicon construction with representation learning based on hierarchical sentiment supervision, in Proceedings of the 2017 Conference on Empirical Methods in Natural Language Processing, Stroudsburg, PA, USA, 2017, Association for Computational Linguistics.
- [136] S. WANG, S. MAZUMDER, B. LIU, M. ZHOU, AND Y. CHANG, Target-sensitive memory networks for aspect sentiment classification, in Proceedings of the 56th Annual Meeting of the Association for Computational Linguistics (Volume 1: Long Papers), Stroudsburg, PA, USA, 2018, Association for Computational Linguistics.
- [137] X. WANG, X. LUO, AND J. CHEN, Social sentiment detection of event via microblog, in 2013 IEEE 16th International Conference on Computational Science and Engineering, IEEE, Dec. 2013.
- [138] Z. WANG, T. ANDERSON, AND L. CHEN, How learners participate in connectivist learning: An analysis of the interaction traces from a cMOOC, Int. Rev. Res. Open Distrib. Learn., 19 (2018).
- [139] B. WEI AND C. PAL, Cross lingual adaptation: An experiment on sentiment classifications, in Proceedings of the ACL 2010 Conference Short Papers, 2010, pp. 258–262.
- [140] J. A. WEI WEIGULLA, Sentiment learning on product reviews via sentiment ontology tree, Acl, (2010).
- [141] A. WEICHSELBRAUN, S. GINDL, AND A. SCHARL, Extracting and grounding contextualized sentiment lexicons, IEEE Intell. Syst., 28 (2013), pp. 39–46.
- [142] A. WEICHSELBRAUN, S. GINDL, AND A. SCHARL, Enriching semantic knowledge bases for opinion mining in big data applications, Knowl. Based Syst., 69 (2014), pp. 78–85.
- [143] J. WIEBE AND E. RILOFF, Creating subjective and objective sentence classifiers from unannotated texts, in Computational Linguistics and Intelligent Text Processing, Lecture notes in computer science, Springer Berlin Heidelberg, Berlin, Heidelberg, 2005, pp. 486–497.
- [144] J. WIEBE AND T. WILSON, Learning to disambiguate potentially subjective expressions, in proceeding of the 6th conference on Natural language learning - COLING-02, Morristown, NJ, USA, 2002, Association for Computational Linguistics.
- [145] J. R. WILLIAMSON, E. GODOY, M. CHA, A. SCHWARZENTRUBER, P. KHORRAMI, Y. GWON, H.-T. KUNG, C. DAGLI, AND T. F. QUATIERI, *Detecting depression using vocal, facial and semantic communication cues*, in Proceedings of the 6th International Workshop on Audio/Visual Emotion Challenge, New York, NY, USA, Oct. 2016, ACM.
- [146] C. WU, Q. XIONG, M. GAO, Q. LI, Y. YU, AND K. WANG, A relative position attention network for aspect-based sentiment analysis, Knowl. Inf. Syst., 63 (2021), pp. 333–347.
- [147] S. WU, I. H. WITTEN, AND M. FRANKEN, Utilizing lexical data from a web-derived corpus to expand productive collocation knowledge, ReCALL, 22 (2010), pp. 83–102.
- [148] D. XU, Z. TIAN, R. LAI, X. KONG, Z. TAN, AND W. SHI, Deep learning based emotion analysis of microblog texts, Inf. Fusion, 64 (2020), pp. 1–11.
- [149] H. XU, B. LIU, L. SHU, AND P. YU, BERT post-training for review reading comprehension and aspect-based sentiment analysis, in Proceedings of the 2019 Conference of the North American Chapter of the Association for Computational Linguistics: Human Language Technologies, Volume 1 (Long and Short Papers), Minneapolis, Minnesota, June 2019, Association for Computational Linguistics, pp. 2324–2335.
- [150] W. XUE AND T. LI, Aspect based sentiment analysis with gated convolutional networks, in Proceedings of the 56th Annual Meeting of the Association for Computational Linguistics (Volume 1: Long Papers), Stroudsburg, PA, USA, 2018, Association for Computational Linguistics.
- [151] A. YADOLLAHI, A. G. SHAHRAKI, AND O. R. ZAIANE, Current state of text sentiment analysis from opinion to emotion mining, ACM Comput. Surv., 50 (2018), pp. 1–33.
- [152] Z. YAN, M. XING, D. ZHANG, B. MA, AND T. WANG, A Context-Dependent sentiment analysis of online product reviews based on dependency relationships, in Icis, 2014, pp. 1–14.
- [153] C.-A. YU, C.-L. TAI, T.-S. CHAN, AND Y.-H. YANG, Modeling multi-way relations with hypergraph embedding, in Proceedings of the 27th ACM International Conference on Information and Knowledge Management, New York, NY, USA, Oct. 2018, ACM.
- [154] H. YU AND V. HATZIVASSILOGLOU, Towards answering opinion questions, in Proceedings of the 2003 conference on Empirical methods in natural language processing -, Morristown, NJ, USA, 2003, Association for Computational Linguistics.
- [155] J. YU, Z.-J. ZHA, M. WANG, AND T.-S. CHUA, Aspect ranking: Identifying important product aspects from online consumer reviews, in Proc. 49th Annu, 2011.
- [156] Y. YU AND X. WANG, World cup 2014 in the twitter world: A big data analysis of sentiments in U.S. sports fans' tweets, Comput. Human Behav., 48 (2015), pp. 392–400.

- [157] D. ZHANG, L. TIAN, M. HONG, F. HAN, Y. REN, AND Y. CHEN, Combining convolution neural network and bidirectional gated recurrent unit for sentence semantic classification, IEEE Access, 6 (2018), pp. 73750–73759.
- [158] J. ZHU, H. WANG, B. K. TSOU, AND M. ZHU, Multi-aspect opinion polling from textual reviews, in Proceedings of the 18th ACM conference on Information and knowledge management, New York, NY, USA, Nov. 2009, ACM.
- [159] L. ZHUANG, F. JING, AND X.-Y. ZHU, Movie review mining and summarization, in Proceedings of the 15th ACM international conference on Information and knowledge management - CIKM '06, New York, New York, USA, 2006, ACM Press.
- [160] H. ZOGAN, I. RAZZAK, S. JAMEEL, AND G. XU, DepressionNet: Learning multi-modalities with user post summarization for depression detection on social media, in Proceedings of the 44th International ACM SIGIR Conference on Research and Development in Information Retrieval, New York, NY, USA, July 2021, ACM.
- [161] M. L. ZOU, M. X. LI, AND V. CHO, Depression and disclosure behavior via social media: A study of university students in china, Heliyon, 6 (2020), p. e03368.
- [162] X. ZOU, J. YANG, AND J. ZHANG, Microblog sentiment analysis using social and topic context, PLoS One, 13 (2018), p. e0191163.
- [163] C. ZUCCO, B. CALABRESE, AND M. CANNATARO, Sentiment analysis and affective computing for depression monitoring, in 2017 IEEE International Conference on Bioinformatics and Biomedicine (BIBM), IEEE, Nov. 2017.

Edited by: Rajni Mohana

Special issue on: Sentiment Analysis and Affective computing in Multimedia Data on Social Network Received: Apr 3, 2023 Accepted: Sep 26, 2023